\documentclass[a4paper, 10pt]{article}
\usepackage{a4wide}
\usepackage{times}
\begin{document}

\title {Software Design and Evolution -- Autumn 2014 Questionnaire}
\date{}

\maketitle

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\paragraph{1. Lectures}
Please indicate for each lecture one of:
(A) very good -- I learned many useful things;
(B) average -- should be improved next time;
or (C) not good -- I learned next to nothing.

Please also enter any brief comments.  (Longer, general comments may be entered on the back of this sheet.)

\begin{table}[h] 
\begin{center} 
\begin{tabular}{|p{0.25cm}p{4.75cm}|p{0.25cm}|p{0.25cm}|p{0.25cm}|p{7cm}|}\hline 
& {\bf Lecture} & {\bf A} & {\bf B} & {\bf C} & {\bf Brief comments} \\ \hline
1	&	Introduction to Software Design and Evolution	 & & & &  \\ \hline
2	&	Smalltalk: A Reflective Language and System	 & & & &  \\ \hline
3	&	Understanding Classes and Metaclasses	 & & & &  \\ \hline
4	&	Reflection and Metaprogramming	 & & & &  \\ \hline
5	&	Introduction to Reverse Engineering / Moose Lab	 & & & &  \\ \hline
6	&	Architectural Extraction	 & & & &  \\ \hline
7	&	Metrics and Problem Detection	 & & & &  \\ \hline
8	&	Software Visualization / Roassal	 & & & &  \\ \hline
9	&	Mining Software Repositories	 & & & &  \\ \hline
10	&	Dynamic Analysis	 & & & &  \\ \hline
11	&	Software Ecosystems	 & & & &  \\ \hline
12	&	Model-Driven Development / Magritte	 & & & &  \\ \hline
\end{tabular} 
\end{center} 
\end{table} 

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\paragraph{2. Exercises}
Please indicate for each exercise one of:
(A) very good -- I learned many useful things;
(B) average -- should be improved next time;
or (C) not good -- I learned next to nothing.

Please also enter any brief comments.  (Longer, general comments may be entered on the back of this sheet.)

\begin{table}[h] 
\begin{center} 
\begin{tabular}{|p{0.25cm}p{4.75cm}|p{0.25cm}|p{0.25cm}|p{0.25cm}|p{7cm}|}\hline 
& {\bf Lecture} & {\bf A} & {\bf B} & {\bf C} & {\bf Brief comments} \\ \hline
1	&	Introduction to Software Design and Evolution	 & & & &  \\ \hline
2	&	Smalltalk: A Reflective Language and System	 & & & &  \\ \hline
3	&	Understanding Classes and Metaclasses	 & & & &  \\ \hline
4	&	Reflection and Metaprogramming	 & & & &  \\ \hline
5	&	Introduction to Reverse Engineering / Moose Lab	 & & & &  \\ \hline
6	&	Architectural Extraction	 & & & &  \\ \hline
7	&	Metrics and Problem Detection	 & & & &  \\ \hline
8	&	Software Visualization / Roassal	 & & & &  \\ \hline
9	&	Mining Software Repositories	 & & & &  \\ \hline
10	&	Dynamic Analysis	 & & & &  \\ \hline
11	&	Software Ecosystems	 & & & &  \\ \hline
12	&	Model-Driven Development / Magritte	& & & & \\ \hline
\end{tabular} 
\end{center} 
\end{table}

\pagebreak


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\paragraph{3. Free-form comments}
Please list any additional comments on any aspect of the course or the exercise sessions.
Of particular interest are:
What parts of the course were good and should not be changed?
What parts were inadequate and should be fixed?
What topics are missing and should be covered in future?
What should we change on this form?

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\end{document}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%